Author's response to reviews

Title: Length of sick leave - Why not ask the sick-listed? Sick-listed individuals predict their length of sick leave more accurately than professionals

Authors:

Nils Fleten (Nils.Fleten@ism.uit.no)
Roar Johnsen (Roar.Johnsen@medisin.ntnu.no)
Olav Helge Forde (Olav.Helge.Forde@ism.uit.no)

Version: 4  Date: 3 October 2004

Author's response to reviews: see over
Dear Editors

MS: 1758495353377884
Length of sick leave - Why not ask the sick-listed? Sick-listed individuals predict their length of sick leave more accurately than professionals.

Many thanks for the response to our manuscript. We have carefully revised the tables to secure that they are divided into cells, and not generated with tabbed text. The margins with black lines are changed from independent lines to margins within the tables. The manuscript is checked again for typos and spelling errors, and some minor correction in punctuation and spelling are done. We hope that the changes made, increases the clarity of the manuscript.

All authors have read and accepted the changes.

Yours sincerely

Nils Fleten